

CrossMark

Citation: Waltmann A, Darcy AW, Harris I, Koepfli C, Lodo J, Vahi V, et al. (2015) High Rates of Asymptomatic, Sub-microscopic *Plasmodium vivax* Infection and Disappearing *Plasmodium falciparum* Malaria in an Area of Low Transmission in Solomon Islands. PLoS Negl Trop Dis 9(5): e0003758. doi:10.1371/journal.pntd.0003758

Editor: Marcelo U. Ferreira, University of Sao Paulo, BRAZIL

Received: November 20, 2014

Accepted: April 13, 2015

Published: May 21, 2015

Copyright: © 2015 Waltmann et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data are available on the DRYAD digital repository. doi:<u>10.5061/dryad.</u> <u>545tv.</u>

Funding: This work was supported by the TransEPI consortium funded by the Bill & Melinda Gates, the National health and Medical Research Council of Australia (NHMRC, #1021544) and the Southwest Pacific International Centre of Excellence in Malaria Research (NIH grant U19AI089686 "Research to control and eliminate malaria in the Southwest Pacific"). This work was made possible through

RESEARCH ARTICLE

High Rates of Asymptomatic, Submicroscopic *Plasmodium vivax* Infection and Disappearing *Plasmodium falciparum* Malaria in an Area of Low Transmission in Solomon Islands

Andreea Waltmann^{1,2}, Andrew W. Darcy³, Ivor Harris⁴, Cristian Koepfli^{1,2}, John Lodo⁵, Ventis Vahi³, David Piziki³, G. Dennis Shanks⁴, Alyssa E. Barry^{1,2}, Maxine Whittaker⁶, James W. Kazura⁷, Ivo Mueller^{1,2,8}*

 The Walter & Eliza Hall Institute, Melbourne, Australia, 2 University of Melbourne, Melbourne, Australia,
National Health Training and Research Institute, Ministry of Health, Honiara, Solomon Islands,
Australian Army Malaria Institute, Brisbane, Australia, 5 National Vector Borne Disease Control Program, Ministry of Health, Honiara, Solomon Islands, 6 School of Population Health, University of Queensland, Brisbane, Australia, 7 Case Western Reserve University, Cleveland, Ohio, United States of America,
Barcelona Centre for International Health Research (CRESIB), Barcelona, Spain

* ivomueller@fastmail.fm

Abstract

Introduction

Solomon Islands is intensifying national efforts to achieve malaria elimination. A long history of indoor spraying with residual insecticides, combined recently with distribution of long lasting insecticidal nets and artemether-lumefantrine therapy, has been implemented in Solomon Islands. The impact of these interventions on local endemicity of *Plasmodium* spp. is unknown.

Methods

In 2012, a cross-sectional survey of 3501 residents of all ages was conducted in Ngella, Central Islands Province, Solomon Islands. Prevalence of *Plasmodium falciparum*, *P. vivax*, *P. ovale* and *P. malariae* was assessed by quantitative PCR (qPCR) and light microscopy (LM). Presence of gametocytes was determined by reverse transcription quantitative PCR (RT-qPCR).

Results

By qPCR, 468 *Plasmodium* spp. infections were detected (prevalence = 13.4%; 463 P. *vivax*, five mixed *P. falciparum/P. vivax*, no *P. ovale* or *P. malariae*) versus 130 by LM (prevalence = 3.7%; 126 *P. vivax*, three *P. falciparum* and one *P. falciparum/P. vivax*). The prevalence of *P. vivax* infection varied significantly among villages (range 3.0-38.5%, *p*<0.001) and across age groups (5.3-25.9%, *p*<0.001). Of 468 *P. vivax* infections, 72.9% were sub-

Victorian State Government Operational Infrastructure Support and Australian Government NHMRC IRIISS. IM is supported by an NHMRC Senior Research Fellowship (#1043345) and AW is supported by an NHMRC Postgraduate Scholarship (#APP1056511). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

microscopic, 84.5% afebrile and 60.0% were both sub-microscopic and afebrile. Local residency, low education level of the household head and living in a household with at least one other *P. vivax* infected individual increased the risk of *P. vivax* infection. Overall, 23.5% of *P. vivax* infections had concurrent gametocytaemia. Of all *P. vivax* positive samples, 29.2% were polyclonal by MS16 and *msp1*F3 genotyping. All five *P. falciparum* infections were detected in residents of the same village, carried the same *msp2* allele and four were positive for *P. falciparum* gametocytes.

Conclusion

P. vivax infection remains endemic in Ngella, with the majority of cases afebrile and below the detection limit of LM. *P. falciparum* has nearly disappeared, but the risk of re-introductions and outbreaks due to travel to nearby islands with higher malaria endemicity remains.

Author Summary

Solomon Islands, an island nation in the Southwest Pacific that has seen dramatic reductions in malaria transmission over the past 20 years, is aiming for malaria elimination. There is an increasing recognition that a substantial reservoir of asymptomatic and often sub-microscopic *Plasmodium spp*. infections exists even in low transmission settings. However, the potential role for these infections in sustaining transmission and the difference in response of the two most common malaria parasites, P. vivax and P. falciparum, to intensified control remains unclear. In May-June 2012, we therefore performed a crosssectional survey of 3501 residents of all ages of Ngella, a low transmission area in Central Islands Province, to assess the prevalence of P. vivax and P. falciparum infection, determine the proportion of sub-microscopic and afebrile infections and evaluate whether gametocytaemic, and thus potentially infectious, individuals are present. Our survey showed a marked epidemiological contrast between P. vivax and P. falciparum. Although prevalence varied significantly among different regions of Ngella, P. vivax remains firmly endemic, with high rates of sub-microscopic, afebrile and genetically diverse infections. The presence of gametocytes among both sub-microscopic and microscopy positive, asymptomatic infections indicates that these infections contribute significantly to sustaining *P. vivax* transmission. *P. falciparum*, on the other hand, appears to be more amenable to control interventions. Only five P. falciparum infected individuals were detected, and all were residents of the same village. These infections carried the same *msp2* clone. This difference highlights the larger challenge of eliminating P. vivax compared to P. falciparum in areas where they are co-endemic. In particular, the challenge posed by the presence of a large reservoir of silent P. vivax infections will need to be addressed if control of this parasite is to be accelerated and elimination achieved.

Introduction

Nations in the Southwest Pacific have endured considerable malaria transmission, with the highest *Plasmodium falciparum* burden outside the African continent and possibly the highest *Plasmodium vivax* transmission in the world [1]. Historically, transmission has ranged from hyperendemic areas in West Papua (Indonesia) and Papua New Guinea [2] to high and

moderate transmission in Solomon Islands and Vanuatu [3], which are the southwestern boundary of global malaria transmission. Intensified control over the last 20 years has resulted in remarkable declines in malaria transmission in this region [3,4], reviving the agenda of elimination. However, it is in these countries where outstanding progress towards elimination has been made, that more knowledge is needed if the vision of malaria elimination is to be realized, such as reliable prevalence estimates, role of low-density, asymptomatic carriers and determinants of transmission maintenance.

In Solomon Islands, the incidence of clinical malaria cases diagnosed by light microscopy (LM) dropped by 90% from 442/1000 population in 1992 [5] to 44/1000 population in 2012 [6]. These drops in incidence are similar to those achieved by the Malaria Eradication Program in Solomon Islands (1970–1975) [7]. National statistics based on passive surveillance indicate that 65% of clinical malaria cases in 2012 were attributable to *P. falciparum*, 33% to *P. vivax* and 2% to mixed *P. falciparum/P. vivax*. Conversely, active case detection surveys indicate that *P. vivax* is the predominant species in the general population [6]. Current malaria transmission appears to be focal, ranging from moderate to high levels in Honiara City (96/1000) and Guadalcanal (64/1000) to very low in Temotu (10.8/1000) and Isabel provinces (1.2/1000).

Temotu and Isabel are the only two provinces in which pilot elimination agenda has been proposed to be actively pursued, having resulted in more intensive control activities and interventions including stratification, active case detection, and the earlier roll out of control activities (e.g. rapid diagnostic tests, RDTs and indoor residual spraying) than the rest of the country [3]. These provinces are also the only areas of Solomon Islands with recent surveys in which both LM and PCR-based diagnoses of *Plasmodium spp*. infections were performed [8,9]. In 2008, a parasite prevalence of 2.7% by LM was found in Temotu, with *P. vivax* accounting for 82.5% of infections. Only 5.5% of these infections were associated with febrile illness. Among a subset of 1,748 samples, which included LM positive, febrile and 10% of LM negative participants, an additional 63 *P. falciparum*, 23 *P. vivax* and 10 mixed *P. falciparum/P. vivax* infections were detected by PCR, indicating a 6.5% prevalence of sub-microscopic infections. Even lower levels of infection were reported in Isabel in 2009: 1 of 8,554 participants had a LM-detectable *P. falciparum* infection (0.01%). In a random subset of 2001 participants, PCR identified an additional 13 (0.55%) *P. vivax* infections.

PCR consistently detects at least twice as many infections as LM [10]. Numerous studies have confirmed that sub-microscopic infections are a common feature of malaria endemic areas, spanning all age groups and involving both *P. falciparum* and *P. vivax* [11–13]. Although these sub-microscopic infections are rarely associated with febrile illness, they have been shown to be efficient gametocyte producers [14–19] and thus constitute a source of ongoing transmission [10].

Given the lack of data from other areas of Solomon Islands, it is currently unknown whether the pattern of asymptomatic, low-density infection carriage identified in Temotu and Isabel [8,9] is unique to these elimination provinces. In addition, whereas these earlier surveys detected a large burden of sub-microscopic infections, they did not determine if these infections were also gametocytaemic and therefore did not assess their potential contribution to transmission. Therefore, we conducted in May-June 2012 a household-based, cross-sectional survey in Ngella, Central Islands Province to determine how common low-density, asymptomatic infections are in communities where transmission is mesoendemic and whether these infections are gametocyte producers and hence, potential contributors to local transmission. This survey is the first epidemiological description of malaria in Ngella since the 1970–1975 Malaria Eradication Program [7] and the only one in Solomon Islands to employ highly sensitive molecular diagnosis for the detection of both blood-stage parasites and gametocytes.

Methods

Ethics statement

This study was approved by The Walter and Eliza Hall Institute Human Research Ethics Committee (HREC number 12/01) and the Solomon Islands National Health Research Ethics Committee (HRC12/022). The informed consent process recognized the community and cultural values of Solomon Islands. Following consultation with and approval by community leaders, community meetings were held to explain the aims, risks and potential benefits of the study. Individual informed consent was obtained from all participants or the parent or legal guardian of children<18 years of age. At the point of collection, all samples were de-identified.

Study site

Ngella, previously known as the Florida Islands, consists of 3 islands, Anchor, Big Ngella and Small Ngella, located approximately 27 miles north of Guadalcanal and 50 miles southwest of Malaita (Fig 1). Along with Tulaghi, Savo, Russel and Buenavista Islands it forms part of the Central Islands Province (Fig 1). Despite their proximity, the three islands of Ngella have diverse geographical characteristics: Anchor Island is characterized by less dense rainforest and sandier soil. Big Ngella is heavily forested, although commercial deforestation is common, and smaller villages are encountered in the Bay area around Tulagi, the provincial capital. The more remote northern villages of Big and Small Ngella and those on the southern coast are larger. The communities of the Utuha Channel lay in an extensive mangrove system and are smaller in size. There is minimal seasonal variation in temperature and despite a northwesterly monsoon from November-April, the distinction between wet and dry season is not pronounced. The most recent census estimates 26,051 inhabitants (approximately 60% of these reside in Ngella), 49% females and a median age of 19.9 years [20]. There is significant migration between Ngella and other malaria endemic areas, in particular Honiara (Guadalcanal) and Malaita provinces. These provinces are well connected to Ngella by a popular ferry service and numerous private, unscheduled motorized boat trips. The Ngella population is serviced by a hospital in Tulagi, six rural health sub-centres and ten nurse aid posts. National malaria statistics describe Ngella as mesoendemic, with a reported Annual Parasite Index [21] of 46.1/1000 in 2012, *P. falciparum* being the main cause of malaria cases [6].

Overall API for Solomon Islands indicates that there were two transmission peaks in 2012 for the months of February and October. As elsewhere in the country, long lasting insecticidal nets and indoor residual spraying are the mainstay of malaria control in Ngella. Cases are diagnosed by LM or RDT and treatment with artemether-lumefantrine has been introduced nationally in 2008.

The last malaria epidemiological report of Ngella [7] described it as 'the most malarious group in all Solomon Islands" and the "most difficult from which to clear malaria". Malariometric surveys preceding the Malaria Eradication Program (March 1965—January 1970, unpublished World Health Organisation Field Reports (reviewed in [8]) identified a combined parasite rate of 69.6% and a spleen rate of 69.3% in the 2–9 years age group. In the same surveys, villages on the North coast had spleen rates in the 80% range and qualified for the hyperendemic classification [7], whereas the villages in the Bay area and South coast were noted to have had spleen rates in the 30–50% range [7].

Study population and blood sample collection

A representative population sample was obtained with a household-based sampling strategy of villages in 5 distinct geographical regions (Fig 1B, Anchor Island, North Coast, Bay, South



Fig 1. Ngella sampling sites and spatial distribution of *P. vivax* **prevalence (qPCR).** The 9 island provinces of SI are shown in the top right inset. A (inset). Central Islands Province. B (inset). Ngella. B. Ngella study catchments and prevalence. Anchor Island (catchments shown in yellow), Bay (catchments in red), Channel (catchments in purple), North Coast (catchments in orange) and South Coast (catchments in blue). The size of the prevalence pie chart reflects village sample size.

doi:10.1371/journal.pntd.0003758.g001

Coast and Channel). The survey included 3501 individuals of all ages \geq 6 months residing in 874 households in 19 randomly selected communities. The households were enumerated and geo-positioned and demographic information of the male and female heads of the household collected. Enumeration, but not geopositioning, was achieved for the households in the villages of the South Coast. The timing of the survey was approximately 4 weeks after the peak of the wet season.

Following consent and enrolment from each participant, a short clinical assessment was conducted (including tympanic temperature, history of fever during the previous 48 hours, history of malaria in the last 2 weeks and spleen size in children 2–9 years old) and demographic information collected (age, sex, residency status and history of travel, bed net use). A febrile participant was defined as having a tympanic temperature \geq 38.0°C and/or a history of febrile illness in the past 48 hours. Study participants who reported being ill at the time of the survey

were diagnosed by RDT (Access Bio, CareStart, USA) and treated if positive with artemether/ lumefantrine, as per the national treatment guidelines. Where available the participant's health records were checked for recent anti-malarial treatment and applicable information recorded.

A 250 µL finger prick blood sample was collected into EDTA-Microtainer tubes (Becton Dickinson, NJ, USA). 50 µL were immediately stabilized in 250 µL RNAProtect (Qiagen, Germany) for RNA studies and stored at ice pack cooling conditions until their transport to a centralized field laboratory. Thick and thin films were prepared for determination of microscopic malaria infection. Haemoglobin measurement was performed with Hemocue HB 301 analyzer. A measurement below 11g/dL was classified as anaemia. Upon return to the centralized laboratory, the RNAProtect fractions were frozen immediately. The remaining 200 µL of whole blood was separated into red blood cells pellets and plasma and promptly frozen.

LM detection of Plasmodium spp. parasites

Giemsa stained blood films were examined under x1000 power. One hundred fields of view were examined before calling a sample "no parasites seen". When a parasite was observed, counts of both white cells and parasites were commenced, and continued until 300 white cells had been counted. The parasite count was then calculated, based on an assumed white cell count of 8,000 white cells/ μ L. However if no further parasites were observed, the process of scanning to a total of 100 fields of view was completed. When only 1 parasite had been observed in 100 fields of view, an assumed count at the notional lower limit of detection of 10 parasites/ μ L was applied, based on a further assumption of an average of 8 white cells per field of view. All slides were stained within 24 hours at the regional malaria laboratory and read by experienced microscopists, all of whom had completed WHO quality assurance courses. All LM positive slides as well as the slides from all PCR positive / LM negative plus 10% of LM & PCR negative slides were re-read by an Australian Level 1 expert microscopist that was blinded to the PCR results. None of the 10% LM negative slides were found to be positive by the expert microscopist. In case of discrepancies between the two microscopy reads, the read of the expert microscopist was considered final.

DNA and RNA extraction

Genomic DNA (gDNA) was isolated from red blood cell pellets (100 μ L, corresponding to 200 μ L whole blood) using FavorPrep 96-well Genomic DNA kit (Favorgen, Taiwan). DNA was eluted in 200 μ L elution buffer and stored at -20°C. The RNA isolation procedure from whole blood in RNAProtect cell reagent has been described elsewhere [22], the only exception being an increased elution volume of 60 μ L of RNase-free water. Due to problems with storage of RNAProtect samples in the field, the quality of the RNA was tested using an RT-qPCR for the human beta globin transcript [23]. This revealed a 10x lower total human RNA concentration than in samples from a comparable study in Papua New Guinea [24]. RNA samples were therefore concentrated 10-fold using a CentriVap Concentrator (Labconco, United States) before testing for the presence of gametocytes.

Molecular detection of Plasmodium spp. parasites

All 3501 DNA samples were first screened using a genus-specific qPCR targeting a conserved region of the 18S rRNA gene [22]. Singleplex species-specific *P. falciparum* and *P. vivax* Taq-man qPCRs and a duplex *P. malariae/P. ovale* qPCR, targeting species-specific regions of 18S rRNA gene, were used to identify species as described previously [22,25]. Prevalence values reported in this study include only those infections confirmed by the species-specific qPCR Taq-man assays. Each detection experiment carried a dilution series of plasmids containing the

target sequence of each PCR (10^4 , 10^3 , 10^2 , 10^1 , 5, 10^0 copies/µL), in duplicate, and were used to determine standard curves and therefore estimate parasite densities (reported as 18S rRNA gene copy numbers/µL). All assays were run in 384-well plate format on the Roche LightCycler480 platform. Those infections detected by qPCR, but not by LM, were defined as sub-microscopic infections. *P. falciparum* and *P. vivax* samples that were positive by species-specific Taqman qPCR were examined for presence of gametocytes using RT-qPCRs targeting the *pfs25* and *pvs25* orthologues, which are expressed only in mature gametocytes, as described previously [22]. All gametocyte assays were also run in 384-well plate format on the Roche LightCycler480 platform.

Plasmodium spp. genetic diversity

All samples that were *P. falciparum* or *P. vivax* positive were genotyped to determine the multiplicity of infection (MOI) using highly diverse size-polymorphic molecular markers *msp2* for *P. falciparum* and *msp1*F3 and MS16 for *P. vivax*, respectively. PCR and capillary electrophoresis were performed with slight modifications to the published protocols [26,27]. Genotyping data was analyzed as described previously [26,27].

Statistical analysis

Study data were collected and managed using REDCap electronic data capture tools hosted at the Walter and Eliza Hall Institute [28]. Analyses were done using the STATA12 statistical software package (College Station, TX). Differences in participant characteristics at enrolment and prevalence differences among geographical areas and groups of individuals were assessed using Chi-square (χ^2) or Fisher's exact tests. Differences in median ages and median household size were explored with quantile regression. Univariable and multivariable logistic regression were used for associations of *P. vivax* infection and exposure variables. Associations with *P. vivax* parasite density were investigated in simple and multivariable linear regression models on only those subjects who tested positive to qPCR diagnosis. Poisson regression analyses were utilized to explore associations between multiplicity of infection and exposure variables.

Results

Study population

A total of 3501 Ngella residents across 874 households were surveyed. The gender and age profiles of the participants were representative of the Central Islands Province population, with 52.5% females and a predominance of younger individuals (median age 18 years). The age distribution was as follows: <2 years, 4.7%; 2–4 years, 10.6%; 5–9 years, 14.8%; 10–14 years, 14.3%; 15–19 years, 7.5%; 20–39 years, 27.0%;>40 years, 21.3%. The majority of participants (95.2%) resided in the village for \geq 2 months. Of 447 participants who spent at least one night outside their village of residence in the last month, 69.1% travelled within Central Islands Province. 73.3% of participants reported having slept under a long lasting insecticidal net the night before and 56.4% owned a bednet for longer than 24 months. Of all households, 84.5% of households reported to have been sprayed with insecticide, and 70.4% of household heads spoke English. Of all participants, 687 (19.4%) had a history of fever in the previous two days, 685 (19.7%) reported feeling unwell/sick at the time of survey and 23.3% had a haemoglobin measurement <11g/dL. No participant aged 2–9 years of age was found to have an enlarged spleen. A detailed description of demographic and clinical characteristics by geographical region is given in <u>S1 Table</u>.

Prevalence of Plasmodium spp. infection by LM

Overall, 130 individuals (3.7%) had *Plasmodium* spp. parasites detectable by LM: 126 *P. vivax*, three *P. falciparum* mono-infections and one *P. vivax/P. falciparum* mixed infection. No infections with *P. malariae* or *P. ovale* were observed. The prevalence of *P. vivax* infection varied significantly by geographical region (p<0.001) (Fig 1B) and was lowest in the South Coast and Anchor regions (0.8%), followed by Channel (3.0%) and Bay (3.5%) and North Coast (11.7%). *P. vivax* prevalence showed strong age trends and peaked in adolescents 10–15 years of age (8.6%, p<0.001) (Fig 2A).

Prevalence of P. vivax by qPCR

Overall, 468 participants (13.4%) had qPCR-detectable infections: 463 were *P. vivax* mono-infections and five were mixed *P. falciparum/P. vivax* infections (0.14%). The 126 *P. vivax* infections and the one mixed infection by LM were confirmed by qPCR. Overall, 72.9% of *P. vivax* infections were sub-microscopic. In two of the catchments, Kelarekeha and Vuturua (Fig 1B), only sub-microscopic infections were observed among 59 and 156 individuals surveyed, respectively. *P. vivax* qPCR prevalence displayed spatial heterogeneity among the five geographical areas and the 19 catchments, varying from 3.0–38.5%. Prevalence by qPCR was highest in villages on the North Coast (25.0–38.5%) and lowest on Anchor Island (3.0–5.5%) (Fig 1B).

Of 874 households sampled across Ngella, 559 had no infected members, 210 had only one infected member and 105 had two or more infected members. There was no association between household size and probability of being infected (p = 0.550). Not taking into account any other variables, there was an increased risk of being infected if at least one other member of the household was infected (OR = 2.59, p<0.001, CI₉₅[2.13, 3.16]).

P. vivax prevalence was age-dependent (p<0.001), lowest in<2 years (3.0%, n = 166) and peaking in the 10–14 year old age group (24.3%, n = 499) (Fig 2A). Prevalence of infection did not differ significantly between male and female participants (p>0.650). Participants who were residents (lived in the village \geq 2 months) were more frequently infected with *P. vivax* than non-residents (infected residents: 13.7% vs. infected non-residents 8.3%, p = 0.045). Once residency status was taken into account, recent travel (defined as spending at least 1 night away from the village of residence in the last month) was not associated with a difference in infection risk (p = 0.300). Those living in a household where the household head speaks English, a proxy for education level, were infected less frequently (12.0%) than those living in a household where the head does not speak English (18.5%, p<0.001). There was a moderate increase in risk of *P. vivax* infection in those who reported not having slept under a net the night before compared to net users (users: 12.6% vs. non-users: 15.6%, p = 0.022).

The majority of *P. vivax*-infected individuals (84.6%) neither reported febrile symptoms (defined as history of fever or measured fever at survey) nor feeling ill (85.4%). Six of the 26 participants that had a measured fever at the time of the survey (tympanic temperature \geq 38°C, 18.8%) were infected with *P. vivax*. Compared to uninfected participants, those with a *P. vivax* infection were less likely to report having had febrile symptoms in the previous two days (uninfected 20.0% vs. infected 15.2%, *p* = 0.014) or report feeling unwell at the time of survey (uninfected 20.5% vs. infected 14.6%, *p* = 0.003). A total of 280 *P. vivax* infections (60.0%) were both asymptomatic and sub-microscopic. There were no significant differences in the proportion of asymptomatic *P. vivax* infected individuals, 19.6% had a haemoglobin<11 g/dL compared to 23.9% of uninfected individuals (*p* = 0.045).



Fig 2. Age trends of *P. vivax* infections. A: *P. vivax* blood stage parasite prevalence by LM and qPCR (error bars represent binomial 95% confidence intervals, Cl₉₅). B: *P. vivax* parasite densities by qPCR (18S DNA copies/l) and LM counts (parasites/l) (error bars represent 95% confidence intervals, Cl₉₅). C: Gametocyte prevalence (in the total sampled population) and positivity (only among *P. vivax* infected) (error bars represent 95% confidence intervals, Cl₉₅). D: *P. vivax* multiplicity of infection (MOI) of blood stage parasites (error bars represent 95% confidence intervals, Cl₉₅).

doi:10.1371/journal.pntd.0003758.g002

Multivariable associations with P. vivax infection

Age was the strongest independent association with *P. vivax*. infection, peaking in 10–14 year olds (Table 1, AOR = 8.41, p<0.001) and remaining relatively steady for subjects aged 15 years and above (Table 1, 15–19 years AOR = 5.03; 20–39 years AOR = 4.75; \geq 40 years AOR = 4.40; p<0.001). The reference group in this analysis was composed of children aged <2 years. Being a local resident, region of residency and living in a household with at least 1 additional infected member were all associated with an excess risk of infection. Significant protective factors included an English-speaking household head and reporting feeling unwell at the time of the survey. Detailed results are given in Table 1.

Table 1. Multivariable associations with P. vivax infection.

		n	PR	AOR	Cl ₉₅	<i>p</i> -value ^a
Region	Bay	1,161	13.7		Reference group	
	Channel	463	13.2	1.01	[0.71, 1.44]	<0.001
	Anchor	857	3.9	0.28	[0.18, 0.44]	
	North Coast	523	31.7	2.52	[1.90, 3.34]	
	South Coast	497	9.9	0.74	[0.52, 1.06]	
Age group (years)	<2	166	3		Reference group	
	2–4	370	5.4	1.90	[0.68, 5.30]	<0.001
	5–9	517	15.1	5.46	[2.13, 14.01]	
	10–14	499	24.3	8.41	[3.31, 21.38]	
	15–19	263	14.5	5.03	[1.88, 13.40]	
	20–39	940	12.6	4.75	[1.88, 12.02]	
	≥40	746	11.8	4.40	[1.72, 11.23]	
Head of the household speaks English	No	661	18.5		Reference group	
	Yes	2,465	12	0.76	[0.50, 0.98]	0.030
Feeling unwell	No	685	14.3		Reference group	
	Yes	2,793	9.9	0.65	[0.48, 0.89]	0.001
Residency (lived in the village for 2 or more months)	No	169	8.3		Reference group	
	Yes	3,317	13.7	1.91	[1.05, 3.48]	0.004
Living in a household with at least 1 other P. vivax carrier	No	2,235	63.8		Reference group	
	Yes	1,266	36.2	1.75	[1.39, 2.20]	<0.001

n = sample size, PR = prevalence of *P. vivax* infection, AOR = adjusted odds ratio (predicted difference in odds of *P. vivax* infection compared to the baseline odds of *P. vivax* infection of the reference group), $CI_{95} = 95\%$ confidence intervals.

^aOverall p-values for variables with multiple levels (age group and region) were derived by use of the likelihood ratio test comparing expanded (includes variable of interest) and nested (does not include variable of interest) models. P-values for the remaining variables are derived from the Wald test.

doi:10.1371/journal.pntd.0003758.t001

P. vivax parasite densities

Parasite densities by LM count were low (estimated geometric mean 24.3 parasites/ μ L, CI₉₅ [19.2, 30.8]) (Fig 2B). Of 130 LM-detectable infections, 55% (n = 70), were at the assumed limit of practical detection, i.e. approximately 10 parasites/ μ L. Similarly, *P. vivax* parasite densities by qPCR were also low (estimated geometric mean 18S DNA copy numbers of 4.6/ μ L, CI₉₅ [3.9, 5.4]) (Fig 2B). In LM and qPCR positive infections, parasite density by LM (parasites/ μ L) and by qPCR (18S DNA copies/ μ L) were correlated (n = 130, R² = 0.76, *p*<0.001). Factors predicting parasite densities by qPCR included age, history of fever in the preceding two days, anaemia (haemoglobin<11 g/dL) and geographical region of sampling. Parasite densities were highest among individuals aged <2 years and decreased in older age groups (*p*<0.001). Detailed results of associations with *P. vivax* density are given in Table 2.

P. vivax genetic diversity

Genotyping results for markers msp1F3 and/or MS16 were obtained from 349 *P. vivax* positive samples. Both markers were highly diverse; 15 msp1F3 alleles and 43 MS16 alleles were detected (Fig 3), resulting in an expected heterozygosity (H_E) of 0.834 for msp1F3 and 0.937 for MS16. Out of 349 samples, 102 (29.2%) carried multi-clonal infections. MOI (combined msp1F3 and MS16), defined as the concurrent infections per individual, ranged from 1 to 4; mean MOI was 1.36. Mean MOI did not differ significantly among age groups (Fig 2D, p = 0.774) or among geographical regions (p = 0.610). *P. vivax*-infected individuals living in a

		n	Geometric mean of parasite density (DNA copies/µl)	Regression Coefficient [§]	Cl ₉₅	<i>p-</i> value ^a	
Region	Вау	1161	2.98	F	Reference group		
	Channel	463	4.73	0.12	[-0.09, 032]	<0.001	
	Anchor	857	4.20	0.12	[-0.14, 0.39]		
	North Coast	523	6.21	0.25	[0.09, 0.40]		
	South Coast	497	7.64	0.45	[0.22, 0.67]		
Age group (years)	<2	166	20.88	Reference group			
	2–4	370	8.99	-0.19	[-0.89, 0.50]	<0.001	
	5–9	517	11.28	-0.1	[-0.75, 0.54]		
	10–14	499	4.97	-0.41	[-1.05, 0.24]		
	15–19	263	3.67	-0.52	[-1.19, 0.15]		
	20–39	940	2.99	-0.62	[-1.26, 0.02]		
	≥40	746	3.03	-0.67	[-1.31, -0.02]		
Fever in the last two days	No	2798	4.39	Reference group			
	Yes	675	6.29	0.2	[0.01, 0.39]	0.036	
Anaemia (haemoglobin<11 g/dL)	No	2682	4.01	Reference group			
	Yes	815	8.46	0.2	[0.03, 0.37]	0.023	

Table 2. Multivariable associations with *P. vivax* density estimates by qPCR.

[§]The regression coefficients represent the predicted changes in 10-fold parasite density (estimated as 18S DNA copies/μL) compared to the mean parasite density of the reference group.

^aOverall p-values for variables with multiple levels (age group and region) were derived by use of the likelihood ratio test comparing expanded (i.e. including the variable of interest) and nested (not including the variable of interest) models. P-values for the remaining variables are derived from the Wald test.

doi:10.1371/journal.pntd.0003758.t002

household with at least one other infected individual had moderately higher mean MOI (1.56) than those who were the sole infected person of the household, but evidence for an association was moderate to weak (mean MOI = 1.42, p = 0.086). Mean MOI was positively associated with qPCR parasite density (p = 0.007).

P. vivax RT-qPCR gametocyte prevalence

In 110 of 468 (23.5%) *P. vivax* infections, gametocytes were detected by the *pvs25* RT-qPCR, resulting in a population gametocyte prevalence of 3.14%. More gametocytes were detected in LM-positive (41.5%) than sub-microscopic infections (16.6%, p<0.001). The proportion gametocyte positive (defined as percentage of gametocytaemic *P. vivax* carriers) was highest in the 2–9 year olds (39.8%, n = 98) and decreasing sharply in the 10–14 year olds (20.7%, n = 121, p>0.001). The lowest proportion gametocyte positive was found among the *P. vivax* carriers in the 15–19 years and>40 years age groups (15.8%, n = 38 and 15.9%, n = 88, respectively) (Fig 2C).





Fig 3. Diversity of *P. vivax* genotypic markers msp1F3 and MS16. Allelic frequencies of *P. vivax* msp1F3, MS16 and the combined msp1F3/MS16 haplotypes. For msp1F3 and MS16, the frequency of the four most common alleles and respective sizes (in basepairs) are shown. The number of unique alleles identified and the estimated expected heterozygosity (H_E) are given below the figure.

doi:10.1371/journal.pntd.0003758.g003

P. falciparum infection

By qPCR, *P. falciparum* was detected in 5 individuals, all of whom were co-infected with *P. vivax*. Of these, only four infections were detectable by LM, one as a co-infection with *P. vivax* and three as *P. falciparum* mono-infections. In two of the LM mono-infections and the mixed infection, only *P. falciparum* gametocytes were observed on the blood smear. The range of parasite densities, by qPCR, was 7.35–364 copy numbers/ μ L and in LM positive samples the densities ranged from 20 to 1430 parasites/ μ L. The low number of *P. falciparum* infections precluded analyses with densities for this species. The presence of gametocytes in the four LM-positive infections was confirmed by *pfs25* RT-qPCR. No gametocytes were detected in the one sub-microscopic *P. falciparum* infection by RT-qPCR.

All five *P. falciparum*-infected individuals resided in the same village (Halavo, circled in Fig <u>1B</u>) and ranged in age from 3 to 60 years. The oldest had a history of fever in the preceding two days. Two of the carriers had a haemoglobin measurement <11g/dL. None of the individuals reported to have slept outside the village in the previous month. All five *P. falciparum* infections were monoclonal and carried the same *msp2* genotype of the Fc27 subtype.

Discussion

Solomon Islands has achieved a remarkable 90% reduction in malaria incidence over the last two decades as a result of scaled-up malaria control interventions [6] and is now intensifying its efforts towards malaria elimination [3]. The present study is the first to undertake sensitive molecular diagnosis at this scale in Solomon Islands and the first large epidemiological description of malaria in Ngella since the Malaria Eradication Program (1970–1975). Our findings illustrate a striking distinction between the epidemiology of *P. falciparum* and *P. vivax* in Ngella.

High prevalence (13.4% by qPCR) and genetic diversity, as well as an increased risk for local residents and evidence of potential within-household transmission indicate considerable levels of endemic *P. vivax* transmission. There was significant variation of *P. vivax* transmission in different regions of Ngella, with the highest prevalence found on the remote North Coast

(25.0–38.5%), which prior to the Malaria Eradication spraying operations was described as holoendemic and having an environment highly favourable to the mosquito [7]. The lowest rates of *P. vivax* infection were observed on Anchor (3.9% by qPCR), where 15 years ago a community-based initiative eliminated a substantial number of breeding sites through environmental management [Lodo, personal communication]. It is therefore likely that the presence of suitable larval habitats and vector abundance may be key factors influencing *P. vivax* transmission on Ngella.

It remains unclear whether autochthonous *P. falciparum* transmission remains in Ngella or parasites are being re-introduced by incoming travelers or returning residents from areas with higher *P. falciparum* burden, such Guadalcanal or Malaita provinces. In this survey, only five *P. falciparum* cases were identified in the village of Halavo (Fig 1B). As all five infections carried the same *msp*2 allele and four were gametocytaemic, a small local outbreak following recent re-introduction seems more likely. This is reminiscent of the situation in epidemic-prone areas of the Papua New Guinea highlands, where a clonal *P. falciparum* epidemic on a background of endemic, low level *P. vivax* transmission has been reported [29]. *P. falciparum* populations in neighbouring Guadalcanal province were in fact found to be of low genetic diversity [30,31]. Based on case statistics at the local health facilities, 30% of malaria cases detected in Central Islands Province are caused by *P. falciparum* [6] indicating that either importation of *P. falciparum* parasites is common or that low levels of endemic *P. falciparum* transmission may remain in some parts of Ngella. Further studies are therefore required to ascertain the absence of endemic *P. falciparum* transmission in this area of Solomon Islands and whether the cases found are the result of inter-island travel.

The current situation of malaria in Ngella (i.e. 3.7% prevalence by LM, clear *P. vivax* dominance and absence of enlarged spleens in children 2–9yrs of age) is a consequence of the dramatic reduction in malaria transmission achieved throughout Solomon Islands in the last 20 years [6]. This change is similar to that encountered at the end of 1974, after approximately 5 years of twice-yearly Malaria Eradication Program spraying. Then, prevalence in 2–9 year olds had dropped from pre-spraying rates of 60% to 1.4% and *P. vivax* became predominant [7]. Similar shifts in malaria epidemiology were also observed in the elimination provinces of Temotu [9] and Isabel [8]. In Temotu, *P. falciparum* accounted for 17.5% of infections in population survey conducted in 2008 [9], but by 2012, the national program's surveillance system reported only *P. vivax* cases from both Temotu and Isabel [6]. This shift in the relative importance of *P. falciparum* and *P. vivax* are not unique to Solomon Islands and have been reported after periods of sustained malaria control from other settings where *P. falciparum* and *P. vivax* occur sympatrically, such as the Amazon [21,32], Central America [4] and Thailand [33].

As in other endemic settings [12,13,34], *P. vivax* infections were of low density and PCR found three times more infections than LM. The majority of infections were not accompanied by febrile symptoms or anaemia. On the contrary, participants who reported feeling unwell or febrile were less likely to be infected with *P. vivax*. While this significantly lower level of febrile symptoms in *P. vivax* carriers is likely to be an artifact of the large samples size it does indicate that *P. vivax* is not a common cause of fever in Ngella. Whereas asymptomatic *P. vivax* infections have been commonly found in areas of high transmission [12,35,36], the advent of molecular diagnosis has revealed that even at low transmission the majority of infections in cross-sectional surveys are symptomless [11,37,38], including in the previous surveys in Temotu [9] and Isabel [8] where 97.1% and 92.9% of *P. vivax* infected individuals infections were asymptomatic, respectively.

Both the presence of *P. vivax* infections and their level of parasitaemia were found to be strongly age-dependent, albeit in different ways: while *P. vivax* parasite densities decreased with age, prevalence of *P. vivax* infections rose throughout childhood and only started

dropping in adolescents and adults. These contrasting patterns are most likely due to local mosquito biting behavior and acquisition of immunity. *Anopheles farauti*, the only coastal malaria vector in Solomon Islands, is biting predominantly in the early evening (i.e. before 10pm) and outdoors [39], when small children tend to be indoors but older ones still active. The increase in prevalence during childhood is thus likely to represent an increase in exposure to infective bites. At all levels of transmission, immunity to *P. vivax* tends to be more rapidly acquired than that to *P. falciparum* [40]. Thus, the strong reduction in prevalence and parasite densities with increasing age in Ngella indicate that *P. vivax* transmission there remains sufficiently high for relatively rapid acquisition of clinical and anti-parasite immunity.

Despite very low overall parasite densities, gametocytes were detected in almost a quarter of all *P. vivax* infections (in 41.5% of LM-positive infections and 16.6% of sub-microscopic infections). Given issues with RNA quality, it is likely that the gametocytaemic reservoir in Ngella was underestimated in our survey and the true prevalence of gametocytes is higher, especially in the sub-microscopic group. Given the rapid and ongoing production of *P. vivax* gametocytes, most if not all, blood stage infections could harbor concurrent gametocytes [41]. Whilst sub-patent *P. falciparum* infections have been shown to infect up to 43.5% of mosquitoes [17,19], the role of sub-microscopic *P. vivax* gametocyte carriage in sustaining transmission is poorly understood. The capacity of sub-microscopic *P. vivax* infections to infect mosquitoes has been established in studies from Thailand [18,42,43], Sri Lanka [44], Peru [45] and malaria therapy settings [14,15], but at varying proportions and with weak associations of gametocyte in endemic settings may mean that the net transmission potential of low-density infections is higher. In Ngella, asymptomatic, sub-microscopic infections of adolescents and adults may thus be an important source of local transmission.

These considerations may constitute a significant challenge to the success of the Solomon Islands malaria control program. The national malaria surveillance system, based on passive case detection and irregular mass blood surveys, only employs traditional microscopy diagnosis. This diagnostic test may not only underestimate the true burden of malaria in the Solomon Islands but also lack the means to detect and attack a substantial part of the *P. vivax* transmission reservoir. Despite outstanding gains in the last two decades, traditional tools of the Solomon Islands malaria control program may therefore have reached their effectiveness in the face of a large and silent reservoir of *P. vivax* infection.

Our observation that people living in a household with another *P. vivax* infected individual is a noteworthy finding. Not only does it indicate likely within-household transmission, but also highlights that reactive case detection strategies [46–48] and focal mass drug administration [34] might be appropriately applied in Solomon Islands. In the Southwest Pacific, MDA campaigns that included primaquine to target the undetectable liver stage parasites have previously been successful in interrupting *P. vivax* transmission on Aneytium Island in Vanuatu [49] and Nissan Island in Papua New Guinea [50]. Combining automated registration of observed cases and rapid identification of transmission foci (e.g. in a spatial decision support system) [51] with reactive mass-screen and treat (MSAT) or with focal, household-based mass drug administration [52,53] should therefore be evaluated as possible additional malaria elimination tools in Solomon Islands and neighbouring Vanuatu. All interventions will be most efficacious if they include routine administration of primaquine to all *P. vivax* infected individuals. This will however require addressing the challenges posed by the potential primaquine toxicity in G6PD deficient individuals.

Supporting Information

S1 Checklist. STROBE checklist. (DOC)

S1 Table. Demographic and clinical characteristics of the study population, by geographical area. (PDF)

Acknowledgments

We would like to extend our sincere gratitude to the communities and community leaders for their support and participation in this study. We acknowledge the indispensable and precious work carried out by the field teams (Freda Pitakaka, Everlyn Darcy, Luito Faarodo, David Ramosala, Leila Tokasi, Anthony Harurihu, Robert Oreitaha, Rebecca Mae, Ethel Lodo, Robert Kali), laboratory officer Rhomson Nuake, vector borne disease control program staff in Honiara (Albino Bobogare, Dr. Lyndes Wini, Erick Hale) and Tulagi (Benjamin Zimbo and Fanny Keru) and Tulagi hospital microscopists (Jentar Saitoha, Stephen Lae, Matthew Mava and George Linton). We also acknowledge Luke Marston for assisting with all aspects of field operations. The support and guidance of Dr. Tanya Russell, Prof. Tom Burkot and Lt. Col. Robert Cooper in Tulagi and the field was indispensable and highly appreciated. We thank Stella Kyvetos and Joan Curtis for ensuring prompt and safe transport of consumables to Solomon Islands and field samples from Solomon Islands to Australia and Dr. Céline Barnadas and Dr. G.L.Abby Harrison for technical assistance in the laboratory. We also thank the manuscript reviewers for their careful assessment and meaningful feedback during the peer review process.

Disclaimer

The opinions expressed are those of the authors and do not necessarily reflect those of the Australian Defence Force.

Author Contributions

Conceived and designed the experiments: AWD GDS MW JWK IM. Performed the experiments: AW AWD CK IH JL VV DP. Analyzed the data: AW CK AEB IM. Contributed reagents/materials/analysis tools: AEB. Wrote the paper: AW CK GDS AEB MW JWK IM.

References

- 1. Muller I, Bockarie M, Alpers M, Smith T. The epidemiology of malaria in Papua New Guinea. Trends Parasitol. 2003; 19(6): 253–9. PMID: <u>12798082</u>
- Battle KE, Gething PW, Elyazar IR, Moyes CL, Sinka ME, Howes RE, et al. The global public health significance of *Plasmodium vivax*. Adv Parasitol. 2012; 80: 1–111. English. doi: <u>10.1016/B978-0-12-397900-1.00001-3</u> PMID: <u>23199486</u>
- Pacific Malaria Initiative Survey Group (PacMISC) on behalf of the Ministries of Health of Vanuatu, Solomon Islands. Malaria on isolated Melanesian islands prior to the initiation of malaria elimination activities. Malar J. 2010; 9: 218. doi: <u>10.1186/1475-2875-9-218</u> PMID: <u>20659316</u>
- 4. World Health Organisation (WHO). World Malaria Report 2013. Geneva, Switzerland: World Health Organisation, 2013.
- Solomon Islands Vector Borne Disease Control Program (SI NVBDCP). Historical Annual Malaria Records. Honiara, Solomon Islands: Ministry of Health.
- 6. Solomon Islands Vector Borne Disease Control Program (SI NVBDCP). Annual Malaria Report 2012. Honiara, Solomon Islands: Ministry of Health, 2013.
- Avery J. The Epidemiology of Disappearing Malaria in the Solomon Islands. Sheffield, United Kingdom: University of Sheffield; 1977. <u>http://ethos.bl.uk/OrderDetails.do?uin = uk.bl.ethos.448417</u>.

- Atkinson JA, Johnson ML, Wijesinghe R, Bobogare A, Losi L, O'Sullivan M, et al. Operational research to inform a sub-national surveillance intervention for malaria elimination in Solomon Islands. Malar J. 2012; 1: 101.
- Harris I, Sharrock WW, Bain LM, Gray KA, Bobogare A, Boaz L, et al. A large proportion of asymptomatic *Plasmodium* infections with low and sub-microscopic parasite densities in the low transmission setting of Temotu Province, Solomon Islands: challenges for malaria diagnostics in an elimination setting. Malar J. 2010; 9:254. doi: 10.1186/1475-2875-9-254 PMID: 20822506
- Okell LC, Bousema T, Griffin JT, Ouedraogo AL, Ghani AC, Drakeley CJ. Factors determining the occurrence of submicroscopic malaria infections and their relevance for control. Nat Commun. 2012; 12 (3):1237.
- Alves FP, Durlacher RR, Menezes MJ, Krieger H, Silva LH, Camargo EP. High prevalence of asymptomatic *Plasmodium vivax* and *Plasmodium falciparum* infections in native Amazonian populations. Am J Trop Med Hyg. 2002 Jun; 66(6):641–8. PMID: 12224567
- Mueller I, Widmer S, Michel D, Maraga S, McNamara DT, Kiniboro B, et al. High sensitivity detection of *Plasmodium* species reveals positive correlations between infections of different species, shifts in age distribution and reduced local variation in Papua New Guinea. Malar J. 2009; 8: 41. doi: <u>10.1186/1475-2875-8-41</u> PMID: <u>19284594</u>
- Steenkeste N, Rogers WO, Okell L, Jeanne I, Incardona S, Duval L, et al. Sub-microscopic malaria cases and mixed malaria infection in a remote area of high malaria endemicity in Rattanakiri province, Cambodia: implication for malaria elimination. Malar J. 2010 Apr 22; 9:108. doi: <u>10.1186/1475-2875-9-108</u> PMID: 20409349
- Boyd MF, Kitchen SF. On the infectiousness of patients infected with Plasmodium vivax and Plasmodium falciparum. Am J Trop Med Hyg. 1937 (17:):253–62.
- Jeffery GM, Eyles DE. Infectivity to Mosquitoes of *Plasmodium falciparum* as Related to Gametocyte Density and Duration of Infection. Am J Trop Med Hyg. 1955; 4(5):781–9. PMID: 13259002
- Nwakanma D, Kheir A, Sowa M, Dunyo S, Jawara M, Pinder M, et al. High gametocyte complexity and mosquito infectivity of *Plasmodium falciparum* in the Gambia. Int J Parasitol. 2008 Feb; 38(2):219–27. PMID: <u>17709108</u>
- Ouedraogo AL, Bousema T, Schneider P, de Vlas SJ, Ilboudo-Sanogo E, Cuzin-Ouattara N, et al. Substantial contribution of submicroscopical *Plasmodium falciparum* gametocyte carriage to the infectious reservoir in an area of seasonal transmission. PLoS One. 2009; 4(12):e8410. doi: <u>10.1371/journal.</u> pone.0008410 PMID: 20027314
- Sattabongkot J, Maneechai N, Rosenberg R. *Plasmodium vivax*: gametocyte infectivity of naturally infected Thai adults. Parasitology. 1991 Feb; 102 Pt 1:27–31.
- Schneider P, Bousema JT, Gouagna LC, Otieno S, van de Vegte-Bolmer M, Omar SA, et al. Submicroscopic *Plasmodium falciparum* gametocyte densities frequently result in mosquito infection. Am J Trop Med Hyg. 2007; 76(3): 470–4. PMID: <u>17360869</u>
- 20. Solomon Islands Government (SIG). 2009. Volume I Report on 2009 Population And HousingCensus: Basic Tables and Census Description. Honiara, Solomon Islands. <u>http://www.mof.gov.sb/Libraries/</u> Statistics/2011_06__Report_on_2009_Population_Housing_Census.sflb.ashx.
- Branch O, Casapia WM, Gamboa DV, Hernandez JN, Alava FF, Roncal N, et al. Clustered local transmission and asymptomatic *Plasmodium falciparum* and *Plasmodium vivax* malaria infections in a recently emerged, hypoendemic Peruvian Amazon community. Malar J. 2005 Dec; 4(6):27.
- Wampfler R, Mwingira F, Javati S, Robinson L, Betuela I, Siba P, et al. Strategies for detection of *Plasmodium* species gametocytes. PLoS One. 2013 27; 8(9):e76316. doi: <u>10.1371/journal.pone.0076316</u> PMID: <u>24312682</u>
- Irenge LM, Robert A, Gala JL. Quantitative assessment of human beta-globin gene expression in vitro by TaqMan real-time reverse transcription-PCR: comparison with competitive reverse transcription-PCR and application to mutations or deletions in noncoding regions. Clin Chem. 2005; 51(12): 2395–6. PMID: 16306108
- 24. Koepfli C, Robinson LJ, Rarau P, Salib M, Sambale N, Wampfler R, et al. Blood-stage parasitaemia and age determine *Plasmodium falciparum* and *P. vivax gametocytaemia* in Papua New Guinea. PLoS One. 2015: in press.
- Rosanas-Urgell A, Mueller D, Betuela I, Barnadas C, Iga J, Zimmerman PA, et al. Comparison of diagnostic methods for the detection and quantification of the four sympatric *Plasmodium* species in field samples from Papua New Guinea. Malar J. 201014; 9: 361. English. doi: <u>10.1186/1475-2875-9-361</u> PMID: <u>21156052</u>
- Falk N, Maire N, Sama W, Owusu-Agyei S, Smith T, Beck HP, et al. Comparison of PCR-RFLP and Genescan-based genotyping for analyzing infection dynamics of *Plasmodium falciparum*. Am J Trop Med Hyg. 2006; 74(6): 944–50. PMID: <u>16760501</u>

- Koepfli C, Ross A, Kiniboro B, Smith TA, Zimmerman PA, Siba P, et al. Multiplicity and diversity of *Plasmodium vivax* infections in a highly endemic region in Papua New Guinea. PLoS Negl Trop Dis. 2011; 5 (12):e1424. doi: 10.1371/journal.pntd.0001424 PMID: 22206027
- Harris PA, Taylor R, Thielke R, Payne J, Gonzalez N, Conde JG. Research electronic data capture (REDCap)-a metadata-driven methodology and workflow process for providing translational research informatics support. J Biomed Inform. 2009; 42(2): 377–81. doi: <u>10.1016/j.jbi.2008.08.010</u> PMID: 18929686
- 29. Mueller I, Kaiok J, Reeder JC, Cortes A. The population structure of *Plasmodium falciparum* and *Plasmodium vivax* during an epidemic of malaria in the Eastern Highlands of Papua New Guinea. Am J Trop Med Hyg. 2002 Nov; 67(5): 459–64. PMID: <u>12479544</u>
- Ballif M, Hii J, Marfurt J, Crameri A, Fafale A, Felger I, et al. Monitoring of malaria parasite resistance to chloroquine and sulphadoxine-pyrimethamine in the Solomon Islands by DNA microarray technology. Malar J. 2010; 9:270. doi: 10.1186/1475-2875-9-270 PMID: 20925934
- Sakihama N, Ohmae H, Bakote'e B, Kawabata M, Hirayama K, Tanabe K. Limited allelic diversity of *Plasmodium falciparum* merozoite surface protein 1 gene from populations in the Solomon Islands. Am J Trop Med Hyg. 2006 Jan; 74(1):31–40. PMID: <u>16407343</u>
- Oliveira-Ferreira J, Lacerda MV, Brasil P, Ladislau JL, Tauil PL, Daniel-Ribeiro CT. Malaria in Brazil: an overview. Malar J. 2010 Apr 30; 9: 115. doi: <u>10.1186/1475-2875-9-115</u> PMID: <u>20433744</u>
- Putaporntip C, Hongsrimuang T, Seethamchai S, Kobasa T, Limkittikul K, Cui L, et al. Differential prevalence of *Plasmodium* infections and cryptic *Plasmodium* knowlesi malaria in humans in Thailand. J Infect Dis. 2009 Apr 15; 199(8):1143–50. doi: <u>10.1086/597414</u> PMID: <u>19284284</u>
- 34. Katsuragawa TH, Gil LH, Tada MS, de Almeida e Silva A, Costa JD, Araujo Mda S, et al. The dynamics of transmission and spatial distribution of malaria in riverside areas of Porto Velho, Rondonia, in the Amazon region of Brazil. Plos One. 2010 Feb 16; 5(2):e9245. doi: <u>10.1371/journal.pone.0009245</u> PMID: <u>20169070</u>
- Pinto J, Sousa CA, Gil V, Ferreira C, Goncalves L, Lopes D, et al. Malaria in Sao Tome and Principe: parasite prevalences and vector densities. Acta Trop. 2000 Sep 18; 76(2):185–93. PMID: 10936578
- 36. Suarez-Mutis MC, Coura JR. [Changes in the epidemiological pattern of malaria in a rural area of the middle Rio Negro, Brazilian Amazon: a retrospective analysis]. Cadernos de saude publica. 2007 Apr; 23(4):795–804. Mudancas no padrao epidemiologico da malaria em area rural do medio Rio Negro, Amazonia brasileira: analise retrospectiva. Portuguese. PMID: <u>17435877</u>
- de AWGWM, Abeyasinghe RR, Premawansa S, Fernando SD. Usefulness of polymerase chain reaction to supplement field microscopy in a pre-selected population with a high probability of malaria infections. Am J Trop Med Hyg. 2011 Jul; 85(1):6–11. doi: <u>10.4269/ajtmh.2011.10-0337</u> PMID: <u>21734117</u>
- Rodulfo H, de Donato M, Quijada I, Pena A. High prevalence of malaria infection in Amazonas State, Venezuela. Rev Inst Med Trop Sao Paulo. 2007 Mar-Apr; 49(2):79–85. PMID: 17505663
- Bugoro H, Hii JL, Butafa C, Iro'ofa C, Apairamo A, Cooper RD, et al. The bionomics of the malaria vector Anopheles farauti in Northern Guadalcanal, Solomon Islands: issues for successful vector control. Malar J. 2014; 13:56. doi: 10.1186/1475-2875-13-56 PMID: 24528850
- Mueller I, Galinski MR, Tsuboi T, Arevalo-Herrera M, Collins WE, King CL. Natural acquisition of immunity to *Plasmodium vivax*: epidemiological observations and potential targets. Adv Parasitol. 2013; 81:77–131. doi: 10.1016/B978-0-12-407826-0.00003-5 PMID: 23384622
- **41.** Bousema T, Drakeley C. Epidemiology and infectivity of *Plasmodium falciparum* and *Plasmodium vivax* gametocytes in relation to malaria control and elimination. Clin Microbiol Rev. 2011 Apr; 24(2):377–410. doi: <u>10.1128/CMR.00051-10</u> PMID: <u>21482730</u>
- 42. Coleman RE, Kumpitak C, Ponlawat A, Maneechai N, Phunkitchar V, Rachapaew N, et al. Infectivity of asymptomatic *Plasmodium*-infected human populations to *Anopheles dirus* mosquitoes in western Thailand. J Med Entomol. 2004 Mar; 41(2):201–8. PMID: <u>15061279</u>
- 43. Sattabongkot J, Maneechai N, Phunkitchar V, Eikarat N, Khuntirat B, Sirichaisinthop J, et al. Comparison of artificial membrane feeding with direct skin feeding to estimate the infectiousness of *Plasmodium vivax* gametocyte carriers to mosquitoes. Am J Trop Med Hyg. 2003 Nov; 69(5):529–35. PMID: 14695091
- 44. Gamagemendis AC, Rajakaruna J, Carter R, Mendis KN. Infectious Reservoir of *Plasmodium vivax* and *Plasmodium falciparum* Malaria in an Endemic Region of Sri-Lanka. Am J Trop Med Hyg. 1991 Oct; 45(4):479–87. PMID: <u>1951856</u>
- **45.** Bharti AR, Chuquiyauri R, Brouwer KC, Stancil J, Lin J, Llanos-Cuentas A, et al. Experimental infection of the neotropical malaria vector *Anopheles darlingi* by human patient-derived *Plasmodium* vivax in the Peruvian Amazon. Am J Trop Med Hyg. 2006 Oct; 75(4):610–6. PMID: <u>17038681</u>

- Rogawski ET, Congpuong K, Sudathip P, Satimai W, Sug-aram R, Aruncharus S, et al. Active case detection with pooled real-time PCR to eliminate malaria in Trat province, Thailand. Am J Trop Med Hyg. 2012 May; 86(5):789–91. doi: 10.4269/ajtmh.2012.11-0617 PMID: 22556075
- Littrell M, Sow GD, Ngom A, Ba M, Mboup BM, Dieye Y, et al. Case investigation and reactive case detection for malaria elimination in northern Senegal. Malar J. 2013; 12:331. doi: <u>10.1186/1475-2875-12-331</u> PMID: <u>24044506</u>
- Sturrock HJ, Novotny JM, Kunene S, Dlamini S, Zulu Z, Cohen JM, et al. Reactive case detection for malaria elimination: real-life experience from an ongoing program in Swaziland. PLoS One. 2013; 8(5): e63830. doi: 10.1371/journal.pone.0063830 PMID: 23700437
- 49. Kaneko A. A community-directed strategy for sustainable malaria elimination on islands: short-term MDA integrated with ITNs and robust surveillance. Acta Trop. 2010 Jun; 114(3):177–83. doi: <u>10.1016/j.</u> <u>actatropica.2010.01.012</u> PMID: <u>20132788</u>
- UCSF GHG. Review of Mass Drug Administration and Primaquine Use, prepared for the Bill and Melinda Gates Foundation. 2014. CA, U.S.A: University of California San Francisco. <u>http://</u> globalhealthsciences.ucsf.edu/sites/default/files/content/ghg/mei-review-of-mda-and-primaquine.pdf.
- Kelly GC, Hale E, Donald W, Batarii W, Bugoro H, Nausien J, et al. A high-resolution geospatial surveillance-response system for malaria elimination in Solomon Islands and Vanuatu. Malar J. 2013; 12:108. doi: <u>10.1186/1475-2875-12-108</u> PMID: <u>23514410</u>
- 52. Shanks GD. Control and elimination of *Plasmodium vivax*. Adv Parasitol. 2012; 80:301–41. doi: <u>10.</u> <u>1016/B978-0-12-397900-1.00006-2</u> PMID: <u>23199491</u>
- 53. Hsiang MS, Hwang J, Tao AR, Liu Y, Bennett A, Shanks GD, et al. Mass drug administration for the control and elimination of *Plasmodium vivax* malaria: an ecological study from Jiangsu province, China. Malar J. 2013; 12:383. doi: 10.1186/1475-2875-12-383 PMID: 24175930